Author’s response to reviews

Title: Differentials of fertility in North and South Gondar zones, Northwest Ethiopia: A comparative cross-sectional study

Authors:

Getu D. Alene (adgetu123@yahoo.com)
Alemayehu Worku (alemayehuwy@yahoo.com)

Version: 6 Date: 27 November 2008

Author’s response to reviews:

We have tried our best to incorporate all comments sent to us by The Biomedical Editorial Production Team. Minor corrections were made on the following sections of the manuscript: Abstract, Introduction, References and Tables. Moreover, qualifications were removed from the manuscript and Email addresses were included as per the direction of the Production Team.